# **ECONSTOR** Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Nshakira-Rukundo, Emmanuel; Mussa, Essa Chanie; Gerber, Nicolas; von Braun, Joachim

# Working Paper Impact of community-based health insurance on child health outcomes: Evidence on stunting from rural Uganda

ZEF Discussion Papers on Development Policy, No. 269

**Provided in Cooperation with:** Zentrum für Entwicklungsforschung / Center for Development Research (ZEF), University of Bonn

*Suggested Citation:* Nshakira-Rukundo, Emmanuel; Mussa, Essa Chanie; Gerber, Nicolas; von Braun, Joachim (2019) : Impact of community-based health insurance on child health outcomes: Evidence on stunting from rural Uganda, ZEF Discussion Papers on Development Policy, No. 269, University of Bonn, Center for Development Research (ZEF), Bonn

This Version is available at: https://hdl.handle.net/10419/204636

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



# WWW.ECONSTOR.EU



# ZEF-Discussion Papers on Development Policy No. 269

Emmanuel Nshakira-Rukundo, Essa Chanie Mussa, Nicolas Gerber, and Joachim von Braun

Impact of Community-Based Health Insurance on Child Health Outcomes: Evidence on Stunting from Rural Uganda

Bonn, January 2019

The **CENTER FOR DEVELOPMENT RESEARCH (ZEF)** was established in 1995 as an international, interdisciplinary research institute at the University of Bonn. Research and teaching at ZEF address political, economic and ecological development problems. ZEF closely cooperates with national and international partners in research and development organizations. For information, see: www.zef.de.

**ZEF** – **Discussion Papers on Development Policy** are intended to stimulate discussion among researchers, practitioners and policy makers on current and emerging development issues. Each paper has been exposed to an internal discussion within the Center for Development Research (ZEF) and an external review. The papers mostly reflect work in progress. The Editorial Committee of the ZEF – DISCUSSION PAPERS ON DEVELOPMENT POLICY includes Joachim von Braun (Chair), Christian Borgemeister, and Eva Youkhana. Chiara Kofol is the Managing Editor of the series.

Emmanuel Nshakira-Rukundo, Essa Chanie Mussa, Nicolas Gerber, and Joachim von Braun, Impact of Community-Based Health Insurance on Child Health Outcomes: Evidence on Stunting from Rural Uganda, ZEF – Discussion Papers on Development Policy No. 269, Center for Development Research, Bonn, January 2019, pp. 39.

ISSN: 1436-9931

Published by: Zentrum für Entwicklungsforschung (ZEF) Center for Development Research Genscherallee 3 D – 53113 Bonn Germany Phone: +49-228-73-1861 Fax: +49-228-73-1869 E-Mail: zef@uni-bonn.de www.zef.de

#### The author[s]:

**Emmanuel Nshakira-Rukundo**, Center for Development Research (ZEF), University of Bonn. Contact: erukundo@uni-bonn.de

Essa Chanie Mussa, Center for Development Research (ZEF), University of Bonn.

Contact: essachanie@gmail.com

Nicolas Gerber, Center for Development Research (ZEF), University of Bonn.

Contact: ngerber@uni-bonn.de

Joachim von Braun, Center for Development Research (ZEF), University of Bonn.

Contact: jvonbraun@uni-bonn.de

# Acknowledgements

We thank Chaira Kofol, Christoph Kronenberg and Andrew Mirelman and the participants of the CINCH-Essen Summer School in Health Economics 2017, African Economic Conference 2017, GLAD Conference Goettingen 2018, for their insightful comments.

## Abstract

While community-based health insurance (CBHI) becomes increasingly integrated into health systems in developing countries, there is still limited research and evidence on its probable health impacts beyond its functions for health financing or for facilitating access to services. Using a cross-sectional data from rural south-west Uganda, we apply a two-stage residual inclusion instrumental variables method to study the impact of community health insurance on stunting in children under five years. Results indicate that each year a household was enrolled in insurance was causally associated with a reduction in the probability of stunting of 5.7 percentage points. Predictive marginal effects show that children in households which have had insurance for at least 5 years had a probability of stunting of only 0.353 compared to 0.531 for children in households with no insurance. Households in CBHI were more likely to attend more free antenatal and postnatal care visits and report fewer illnesses and reported less health expenditures. Moreover, CBHI enrolment was also associated with reduced health costs. We recommend that developing countries should facilitate the expansion of community health insurance scheme not only for their contribution to health financing but even more for mortality and morbidity aversion.

Keywords: Community-based health insurance, Child Stunting, Two-Stage Residual Inclusion, Rural Uganda

JEL Codes: I130, I150, I100, I555

# 1 Introduction

Community-Based Health Insurance (hereafter CBHI) schemes are a particular form of health insurance systems emanating from community social support systems (Criel, Atim, Basaza, Blaise, & Waelkens, 2004), often working in the rural informal sector, and operating without profit motivations (Bennett, Creese, & Monasch, 1998). The schemes evolved in the 1980s especially in resource-poor developing countries where tax-funded and other health insurance platforms were non-existent (Carrin, Waelkens, & Criel, 2005; Ekman, 2004). Over the last few decades, CBHI has evolved as an essential buffer for financial protection and enabling access to health services, especially in poor rural communities. They currently form essential building blocks to achieving universal health coverage in developing countries (Wang & Pielemeier, 2012) and have been adopted across many countries with varying forms and levels of government involvement.

Over the last couple of years, there has been an increasing number in studies on CBHI, with significant focus on enrolment and drop out (Atinga, Abiiro, & Kuganab-Lem, 2015; De Allegri et al., 2006; Dror et al., 2016; Panda, Chakraborty, Dror, & Bedi, 2014), facilitating access to health services (Jutting, 2004; Mebratie, Sparrow, Alemu, & Bedi, 2013; Smith & Sulzbach, 2008; Sood & Wagner, 2016), financial protection (C. V. Nguyen, 2012; H. T. H. Nguyen, Rajkotia, & Wang, 2011; Sepehri, 2014; Sepehri, Sarma, & Oguzoglu, 2011), and wider welfare as well as economy-wide impacts on risk-coping and managing shocks (Asfaw & von Braun, 2004a, 2004b; Landmann & Frölich, 2015; Yilma et al., 2014, 2015). However, evidence remains thin on the impacts of CBHI on the health indicators of the insured, especially children. For instance, of the several systematic reviews in recent years (Adebayo et al., 2015; Dror et al., 2016; Ekman, 2004; Mebratie et al., 2013; Reshmi, Sreekumaran, & Unnikrishnan, 2016; Spaan et al., 2012)), only one, (Acharya et al., 2013) reported some impacts on health outcomes, in six of the nineteen papers they review. Acharya et al. (2013) found this surprising, that research had paid limited attention to understanding if health insurance had any effects on health

outcomes of the insured in developing countries. This study, therefore, seeks to respond to this research gap.

On the other hand, stunting remains a major problem to the health of children in developing countries. Low height for age affects an estimated 165 - 170 million children in the world and affects developing countries disproportionately (Prendergast & Humphrey, 2014; Stevens et al., 2012). The effects are not only detrimental to a child's young life but also sustain into adulthood, affecting educational, health, employment and cognitive abilities later in life (Case & Paxson, 2010; Dewey & Begum, 2011; Glewwe, Jacoby, & King, 2001; Vogl, 2014). Thought stunting has reduced in the last couple of years, an estimated 29 percent of under-5 children were stunted in 2016 (UBOS & ICF International, 2012). Extrapolating from the 2014 census (17.7% of 34.8 million under 5 years), indicated that about 1.8 million children were stunted in 2014. Nutritionists have warned that if the current status of stunting is not improved, stunting could lead to loss of more than half a million lives between 2013 and 2025 (Namugunya et al., 2014).

The main aim of this paper is to therefore combine these two issues, on the one hand and opportunity of the growth of insurance platforms in developing countries, in Uganda in particular and on the other a long-standing child health challenge. This paper uses cross-sectional data from south-western Uganda, collected in an area with a large CBHI scheme and studies the effect of CBHI participation on child stunting. One of the challenges of studying the effect of health insurance on health outcomes is the endogeneity between health insurance and health outcomes (Levy & Meltzer, 2008). To account for such endogeneity the paper applies a novel Instrumental Variable (IV) approach, the Two-Stage Residual Inclusion (2SRI) approach (Terza, Basu, & Rathouz, 2008; Wooldridge, 2015). Using the framework outlined by Terza (2017), the method facilitates not only the analysis of the decision to enrol in CBHI but also incorporates CBHI intensity, measured by the number of years a household participated in CBHI.

This paper's major contribution to the health economics literature is the demonstration that CBHI can influence health dimensions beyond its primary goals of resource mobilisation for health systems and financial protection for households in developing countries. The results indicate that the probability of child stunting reduced by 5.7 percentage points for each year a household was enrolled in CBHI. Predictive marginal effects reflect that contrasted with children in households not enrolled in CBHI whose probability of stunting was 51.5 percent, probability of stunting for children in households in CBHI reduces from 49.2 percent with CBHI membership for only one year to 37.6 percent if they remained members for 5 years. By asking and answering these questions, some more evidence that health of children improves when their households enrol in CBHI is provided.

The rest of this paper is organised as follows. Section 2 positions this paper on why stunting an avoidable child health problem in Uganda. Section 3 reviews the current literature on the effects of CBHI and other voluntary health insurance programs on health outcomes. Section 4 gives the empirical strategy of the paper by elaborating on the data and instrumental variable identification strategy. Section 5 provides the detailed results and section 6 gives a discussion and concludes the paper.

# 2 Health insurance and health outcomes in developing countries

Our work adds to a just emerging body of research in response to this research gap on health insurance and health outcomes. Evidence is still thin and largely mixed. A couple of papers find positive effects. For instance, in Nigeria, (Hendriks et al., 2016, 2014) studied the impact of health insurance on hypertension over the short and medium term. In the first paper, (Hendriks et al., 2014), they find that systolic blood pressure reduced by 5.2 mm Hg more in villages with CBHI while diastolic blood pressure reduced twice as much compared to villages with no CBHI. After 5 years of the CBHI intervention, Hendriks et al. (2016) find sustained effects indicating that systolic blood pressure maintained a greater reduction of 4.97 mm Hg in CBHI villages. Sood and Wagner (2016) found that insurance had positive effects on post-operation recovery self-assessed measurements. Wang, Yip, Zhang, and Hsiao (2009) found overall improvements on the health status of the insured in rural China, while Pan, Lei, and Liu (2016) also report improved self-reported health for those enrolled in the urban insurance program in China.

However, a couple of other papers find no evidence of any effect. While studying the effect of health insurance expansion on child mortality in Costa Rica, Dow and Schmeer (2003) find very limited evidence on this causal effect. In a similar manner, Thornton et al. (2010) do not find significant effects in Nicaragua. Likewise, in China, Lei and Lin (2009) do not find any improvement in self-reported health status of the insured, while Sood et al. (2014) do not find any improvement in mortality of the insured in India. Probably a more extreme finding is Fink, Robyn, Sié, and Sauerborn (2013) who find that health insurance in Burkina Faso was associated with increased mortality of the older enrolees, a result they attribute to the declining quality of care.

A couple of studies that look into child health outcomes also provide mixed results. Fink et al. (2013) did not find any effects on child health measured by mortality. However, three studies find positive effects on child anthropometrics. Lu et al. (2016) used demographic and health surveys data, and health facilities data, and found that enrolment in Rwandan CBHI was associated with reducing the probability of stunting for children between 6 and 24 months by 14 percentage points. Quimbo, Peabody, Shimkhada, Florentino, and Solon (2011) found that voluntary insurance had a causal effect of up to 12 percentage points reduction in wasting in the Philippines while Schoeps et al. (2015) found that the risk of mortality was 46% lower for children in insurance compared to their uninsured counterparts in Burkina Faso.

There are a couple of differences between this paper and these studies on child health and new dimensions that this paper contributes to the literature. Two issues are in relation to (Lu et al., 2016) study in Rwanda. Whilst Rwanda provides one of the best examples of CBHI in scale-up, it can hardly be considered as voluntary health insurance, and therefore diverges from one of the key precepts of CBHI - voluntary enrolment. Since 2007, CBHI in Rwanda has been compulsory (Nyandekwe, Nzayirambaho, & Kakoma, 2014) and the health insurance law was further revised in 2015 (Government of Rwanda, 2016). CBHI in Uganda, on the other hand, is completely voluntary and outside the government public health sector services. Secondly, while Lu et al. (2016) have access to rich demographic and health survey data, they chose a narrow 6-24 months population. We enhance on this by widening the age group to all under-fives and not a section of them. The Philippines study (Quimbo et al., 2011) uses is of limited application to many sub-Saharan African countries because of the lack of clinical data. While clinical data might be more precise in the measurement of health outcomes, in countries like Uganda it would miss a large sub-population with limited access to health facilities. For instance, only 58 percent of mothers delivered in health facilities in Uganda (UBOS & ICF International, 2012). This survey was therefore designed with this in mind and hence captures all households at both ends of health facility access and utilisation.

## 3 The empirical strategy

#### 3.1 The Data

Between August 2015 and April 2016, we carried out a household survey in southwestern Uganda in two districts in which the Kisiizi hospital CBHI scheme operates. The Kisiizi CBHI scheme is one of the 21 and the biggest CBHI scheme in Uganda, providing insurance coverage to close to 42,000 individuals. The scheme is a provider-based scheme, run by a rural hospital. At the time of this study, households paid premiums ranging from Uganda shillings 10,000 (equivalent to approximately US\$3) per person for households of up to 11 people to Uganda shillings 28,000 (approximately US\$8) for households of 2 people.<sup>1</sup>

Membership in insurance is based on group such that for a household to be insured, it has to be part of group. Most of the groups are pre-existing informal funeral insurance groups, which are integral in the promotion of different kinds of insurance(De Weerdt, Dercon, Bold, & Pankhurst, 2007; Dercon, De Weerdt, Bold, & Pankhurst, 2006; Dercon, Vargas, Clarke, Outes-Leon, & Taffesse, 2014). For a funeral insurance group to enrol its members, two conditions have to be met. The first is that the group has to have at least 30 households. For these smaller groups, all the households have to subscribe. The second condition pertains to larger groups. In these larger groups, some with more than 100 households, at least 50% of households have to subscribe and all members of the subscribing households are required to enrol. However, while enrolment is group based, the scheme is not a typical group insurance. There is no joint liability within the groups and households enrol at an individual basis with individual premiums. Households enrol as a full unit, such that, in principle, when a household enrols, all members of the household are enroled and once some members are drop out, the whole household drops out. Another important feature of this scheme is the timing of coverage. In order to further control moral hazard and adverse selection, members are fully covered once they have been in insurance for more than one year. Once a household member is ill with in only one year of enrolling, insurance covers only 10% of the cost. These kinds of strategies to limit adverse selection have been applied in other community insurance scheme. For instance in a Nigerian scheme, enrolled people wait for about 36 days before they can be covered by insurance (Bonfrer, Van De Poel, Gustafsson-Wright, & van Doorslaer, 2015). In the case of this scheme, a waiting time of one year is certainly more conservative. Typically, insurance covers basic primary care, maternity

<sup>&</sup>lt;sup>1</sup>Average annual exchange rate of US\$ 1=3400 UGX in 2015. According to the 2012 National Household Survey (UBOS, 2014) average annual household incomes for Kigezi region were Uganda shillings 4.1 million, derived from reported monthly income of Uganda shillings 343,000. This implied that total premiums for a household of 11 members would be 2.4 percent of total annual household income and 1.4 percent of household income for a household of 2 members. Using the 2016 survey (UBOS, 2017), annual premiums are equivalent to 1.8 percent of average annual household incomes for a 11 member household and 1 percent for a 2 member household

care, surgeries, and outpatient and inpatient services. Outpatient services for chronic illnesses and substance abuse related illnesses and injuries are excluded.

In partnership with this scheme, we used a multi-stage simple random sampling, and surveyed 464 households in fourteen (14) villages in two (2) districts. The survey modules included a household demographic module which collected information on household occupancy; a child and maternal health module which collected information on health care seeking behaviour for mothers and children; a nutrition module which collected information on household food availability and intake. The survey collected detailed information on household social and economic welfare using durable assets holdings and other endowments in agriculture, water and sanitation, and housing. The health insurance and social connectivity modules collected information regarding household insurance status, group membership and participation, and knowledge of insurance such as premiums and benefit package. In line with emerging tools for understanding enrolment in community insurance in sub-Saharan Africa, the survey incorporated a detailed module on perceptions about several aspects of health insurance. In order to get a probably better understanding of health insurance, we also collect birth registry data recorded in birth registers in health facilities with a functioning maternity facilities. Within our region of data collection, data is collected from six health facilities. We present this data as part of summary statistics and use only household data in the casual estimations.

Ethical approvals for data collection was conducted by the Mengo Hospital Research and Ethics Review Committee and an ethical certificated was provided by the Uganda National Council of Science and Technology (Reference Number SS-39369). Further ethical reviews were undertaken by the Centre for Development Research, University of Bonn research committee and the local research partner, Kisiizi Hospital Research Committee. Verbal consent was obtained from the district administration health teams, local government authorities, village leaders and respondents.

# 3.2 Identification strategy: Two-Stage Residual Inclusion Instrumental Variable method

A lot of literature attests to the fact that insurance increases health care utilisation because it removes financial barriers to access. However, in some instances, utilisation of care might lead to over consumption of care which might in turn result in spending of health, hence defeating the financial protection purpose (Wagstaff & Lindelow, 2008). But irrespective of the financial protection situation after insurance, we take that consumption of more health services should, in principle, lead to improved health. But such a causal relationship is difficult to establish because of endogeneity between health insurance and the health outcomes of the insured people (Levy & Meltzer, 2004, 2008). Our identification strategy will therefore utilise an instrumental variable approach to overcome endogeneity problems (Angrist & Imbens, 1995; Angrist, Imbens, & Rubin, 1996). Instead of the conventional Two-Stage Least Squares (TSLS) instrumental variable approach, we employed a Two-Stage Residual Inclusion (2SRI). The 2SRI method, also called the Control Function method introduces extra regressors which break the correlation between the endogenous explanatory variable and the unobservables affecting the outcome (Wooldridge, 2010, p. 126). A couple of reasons are provides as to why the 2SRI method is more appropriate for a study like ours (Cai, Small, & Have, 2011; Terza et al., 2008; Wooldridge, 2010). First, it provides a straight forward way of testing the hypothesis that the endogenous explanatory variable is actually endogenous buy looking at the behaviour of the residuals. Secondly, the model is robust in addressing endogeneity in health economics studies of non-linear relationship that the conventional two-stage least squares and two-stage predictor substitution methods.

Moreover, our treatment variable measured two related parts; namely the treatment and the treatment intensity, we used an alternative estimation of the 2SRI which combines these two parts, following Terza (2017) Stata code implementation guidelines. It is a common occurrence that when treatment is offered, treatment intensity varies across the treated (Angrist & Imbens, 1995) but also the proportion of non-compliers is large and non-ignorable (Mullahy, 1998). In a case like this, Mullahy (1998) proposes a twopart model that assumes that the probability of the treatment, Pr(y>0|x) is governed by a probability model like a Probit in the first part and that E[ln(y)|y>0, x] and is a linear function of x, e.g.,  $E[ln(y)|y>0, x] = x\beta$  (part two).

These two steps produce both the probability of the treatment and the predicted treatment intensity. Residuals are then manually generated and included in the second stage outcome model. Using Mullahy's smoking and birthweight data (see (Mullahy, 1997, 1998)), Terza (2017) employs this version of a non-linear instrumental variable model to analyse of the effect of maternal smoking on birthweight. One other convenience of a 2SRI model is the ease with which the tests of endogeneity and endogeneity control are done. It is important to make sure that the first stage F-statistic for the joint significance of the instruments meets the threshold level of 10 (Staiger & Stock, 1997; Stock, Wright, & Yogo, 2002; Stock & Yogo, 2005). In addition, when the included residuals are significant, it shows that endogeneity was indeed present and also well controlled for in the model (Gibson et al., 2010; Pizer, 2009; Staub, 2009).

Just like the maternal smoking example in Mullahy (1997), a large part of our sample were not enrolled in insurance (x = 0) and in addition, those whose treatment was taken have varying treatment intensities such that x|x>1 = 1, 2... This, therefore, required a two part-part instrumental variable Probit model (Mullahy, 1998). To model the Two Stage Residual Inclusion (2SRI) model, we the guidelines detailed in Terza (2017). In the first part of the two-part first stage of the 2SRI, we estimated the probability of participating in CBHI ( $\alpha_1$ ) by regressing the instruments and other covariates using Probit model.

CBHI Status<sub>i</sub> = 
$$\beta_0 + \beta_1 Z_i + \beta_2 X_i + \epsilon_i$$
 (1)

Where CBHI Status<sub>*i*</sub> is the treatment dummy corresponding with 1 if household *i* participated in CBHI and 0 otherwise,  $Z_i$  is a vector of the instruments used,  $X_i$  stands

for a vector of a child, household and village covariates included in the model, and  $\epsilon$  is the error term.

Using this model in this first step of the first stage, we predict and stored the results for the probability of being insured. Let us call this probability  $\alpha_1$ . In the second part of the first stage, we fit a Generalised Linear Model (GLM) on the treated (insured) sub-sample whose dependent variable is treatment intensity measured by the number of years of continuous insurance. This GLM model is specified with a Gaussian distribution and the default link identity, specifications for continuous outcomes.

CBHI Years<sub>i</sub> = 
$$\beta_0 + \beta_1 Z_i + \beta_2 X_i + \epsilon_i$$
 (2)

Similarly, in this equation, we include in all the instrumental variables and all other covariates included in the model. From this second step of the first stage, we predict and store the mean years of insurance. We call this  $\alpha_2$ . The residuals  $\eta_i$ , are derived as the differences between the observed insurance intensity and the product of predicted insurance status and predicted mean intensity.

$$\eta_i = \text{CBHI Years}_i - \alpha_1 * \alpha_2 \tag{3}$$

The second stage of the 2SRI model, we fit a Probit model of the outcome (stunting prevalence) on insurance intensity all other covariates and the residual inclusion estimator.

Stunted<sub>i</sub> = 
$$\beta_0 + \beta_1 \text{CBHI Years}_i + \beta_2 X_i + \beta_3 \eta_i + \epsilon_i^{2SRI}$$
 (4)

We then reported average partial effects of the probit model of outcome (margins) to show the effect of insurance on child stunting. In these models, we controlled for individual child-specific variables such as child age, sex, if they took a vitamin A supplement in the last months, immunisations taken and others. We also controlled for household-specific variables such as education of the parents, household durable assets, agricultural assets and use of protected water sources. We further included a perception index developed from a principle components analysis that takes a single principle component from several dimensions of perceptions concerning health insurance. We then included several village level controls. Because we generate the residuals from the two first step stages, bootstrapping is recommended to obtain the correct standard errors (Terza, 2017; Wooldridge, 2010). We bootstrap up to 1,000 replications for all the estimated models.

#### 3.3 The instrumental variables

The standard validity of our instruments is hinged on meeting two basic conditions, that; (1) the instruments have a strong predictive association with the treatment (CBHI status) and, (2) that they do not have any significant association with our outcome (stunting prevalence), such that the only effect of the instruments on the outcome is through the treatment. In this regard, we adopt three instruments, namely: (1) village CBHI demand rate (2) size of a burial group a household belonged to and (3) leader and social influence experienced by a household on the decision to enrol in CBHI. Apart from the challenges of finding one very strong instrument, using three instruments also helps in model efficiency gains (Chao & Swanson, 2005; Hansen, Hausman, & Newey, 2008; Roodman, 2009). We expound on each of them.

Latent Village CBHI demand rate: a common IV in health economics studies is aggregated prevalence of a treatment at community or sub-national level. For instance, Stukel et al. (2007) used regional catheterisation rates as an instrument for a patient's probability of receiving a cardiac catheterisation treatment while Jung and Streeter (2015) used the fraction of individual enrolled in insurance at the community level as an instrument for enrolling in insurance in China. The logic in this type of instrument is that the probability of an individual receiving a treatment is correlated with the rate at which the treatment is available in the locality of the individual.

The intuition in which village latent CBHI demand rate is constructed is slightly different. All the none-CBHI households were asked a hypothetical question to state their preference in joining a village group they did not already belong to, to capture their latent preference to join CBHI. While moving from one burial group to another is very restrictive, it is possible when leaders agree to the move and only when the groups belong to the same kin association. Expressing a wish to join a burial group that participated in CBHI showed a household's latent demand. Village CBHI latent demand rate was therefore constructed as the ratio of uninsured households who wish to join a CBHI-participating burial group in a village.

$$Village CBHI demand rate = \frac{Non CBHI households wishing to join a CBHI burial group}{Total non CBHI households}$$
(5)

The demand rate ranged from 0 to 75 percent.

Size of the burial group: Membership in a burial group was one of the basic requirements of enrolling in CBHI but the size of the burial group in which a household belonged to also mattered. Earlier research in this area showed that over 96 percent of the households belonged to a burial group (Musau, 1999) and in this data, all respondents belonged to one. In societies with egalitarian and mutual assistance traits in regards to risk sharing and mutual insurance, group sizes matter (De Weerdt & Dercon, 2006; Dercon et al., 2006). One way in which group size is important is that larger groups reduce the possible egalitarian preferences possibly because of reduced cohesion between the group members and increase free riding (Stahl & Haruvy, 2006). But also, the larger the group the higher the per capita utility from risk sharing (Genicot & Ray, 2003) and therefore the reduced need for taking on other formal insurance<sup>2</sup>. There are a number of ways in which the group size is expected to influence enrolment decisions. As we mention earlier, requirements set by the scheme that pertain to groups affect which groups can have their members enrolled and those who may not. Whilst preference might be given to larger groups because more members can monitor each other for adherence (Carpenter, 2007), smaller groups have more cohesion (Stahl & Haruvy, 2006) and therefore easy to manage. We, therefore, expect a negative association between group size and enrolment status.

Leader and Social influence: The third instrument is influence from leaders and other social contacts that may be experienced by households in the decision to enrol in CBHI. Since household enrolment was group facilitated, it was expected that group leaders, often clan elders, have a significant role they play in influencing households to enrol. Social influence was there assessed as a composite perception indicator, following previous studies that show that perceptions regarding social influence were important for household enrolment in insurance (Jehu-Appiah, Aryeetey, Agyepong, Spaan, & Baltussen, 2012). Respondents were asked five statements with Likert Scale responses

<sup>&</sup>lt;sup>2</sup>We are also aware of literature with a neutral take on the size of the group as regarding to group contributions, such as Fafchamps and La Ferrara (2012). However, one major distinction with Fafchamps and La Ferrara (2012) is the urban nature of his sample. Conventionally, urban residents often have multiple social safety nets and have better access to these safety nets - such as financial services. Therefore group sizes might matter more for rural residents than it does for urban residents.

ranging from 1 to 5; where 1 corresponded with "do not agree" and 5 corresponded with "completely agree".<sup>3</sup> Using Principal Components Analysis, these responses were reduced to one composite indicator of leader and social influence. (Chatterji, 2006) also used a perception instrument to study the effect of illicit drug use during high school on educational attainment.

The validity of these instruments is hinged on how they meet the relevance condition and the exclusion restriction. Table 1 below, show the first stage and reduced form associations of the IVs with CBHI status and stunting respectively. All the three instruments had a strong individual association with CBHI status and their combination shows that this strong association is maintained as indicated in Model 1 of Table 1.

	(1) CBHI Status		(2) Stunting	
	Coeff	SE	Coeff	SE
Village CBHI demand rate	0.801	(0.620)	0.628	(0.459)
Social Influence	0.159**	(0.068)	-0.009	(0.043)
Size of the burial group	-0.018***	(0.004)	0.001	(0.003)
Constant	-0.046	(1.319)	-0.575	(0.886)
All other covariates	YES		YES	
N	464		464	

 Table 1: First stage and reduced form associations

Robust standard errors in parentheses \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

Given that we use more than one instrument, it is always advisable to check that none of the instrumental variables are correlated with the structural error term. Normally, the over-identification can be performed in a straight forward manner in other

<sup>&</sup>lt;sup>3</sup>The five questions are (1) We learn from out neighbours about the things we do such as which community groups to join (2) Village opinion leaders influence us about the programmes we enrol in, such as insurance (3) Our friends and other extended family members influence our decision to enrol in insurance (4) Enroling in insurance as an individual household would be better than the current condition to belong in a group (5) The experiences of other community members with insurance affects our decision to enrol in insurance

variations of the instrumental variable methods (estimating the Sagan or Hansen tests) but for our multi-step approach, we use a regression based approach that follows Wooldridge (2010, p 136-137). We estimate the equation:

$$\eta_i = \beta_1 Z_{ij} + \beta_2 X_{ij} + \epsilon \tag{6}$$

Where  $\eta_i$  are the first stage residuals,  $Z_{ij}$ 's and  $X_{ij}$ 's are the instruments and covariates respectively. From this equation we can obtain the overidentification test statistic which is N (our sample) times the R-squared, given as  $nR^2$ . Our over-identification test statistic is 3.898 with an associated p-value of 1.000 and we can comfortably say that our instrumental variable model does not reject the over-identification restrictions.

This causal relationship can be illustrated using Direct Acyclic Graphs (DAGs) (Stanghellini, 2004; Textor, van der Zander, Gilthorpe, Liśkiewicz, & Ellison, 2017), which illustrates that the IVs are perfect predictors of CBHI status, hence the causal link from the exposure to the outcome. It also further illustrates that there is no relationship between the IVs and stunting.



Figure 1: DAG for causal association of CBHI membership and stunting

# 4 **Results**

#### 4.1 Descriptive results

Stunting is assessed using the World Health Organisation growth monitoring standards as a height-for-age Z-score with a standard deviation of -2 or less (Duggan, 2010). Overall, 42.2 percent of the children in the sample were stunted. Children in the lowest wealth quintile had a higher prevalence, of 58.5 percent while only 30.2 percent of children in the highest wealth quintile were stunted.



Our outcome of interest is stunting prevalence. Overall, 42.2 percent of the children in our sample had a height-for-age z-score less than -2 standard deviations, implying that they were stunted. Stunting prevalence was similar to the average in south western Uganda, 41.7 percent but higher than the national average of 33.1 percent as assessed by the 2011 Demographic and Health Survey (UBOS & ICF International, 2012). In terms of the distribution of stunting, children in the lowest socioeconomic welfare index had a higher prevalence of stunting, just below 60 percent while only 30 percent of children in the highest wealth quintile were stunted.

43.8 percent of the households surveyed were enrolled in CBHI at the time of the survey. This was found to be substantially higher than previously reported insurance coverage of 15 percent in the same area (Dekker & Wilms, 2010) or 30 percent assumed

by (Twikirize & O'Brien, 2012) indicating significant improvements in coverage. For CBHI participating households, the average length of continuous participation was 5.2 years. The distribution of years in CBHI was right-side calibrates to a maximum of 11 years because of a sparse distribution after 10 years. In 30.4% of the households, at least one of the parents had some secondary level education. The average age of mothers interviewed was 30.2 years while the average age of the children whose data was recorded was 30.2 months. Some 63.4% of the mothers had exclusively breastfed their youngest child for up to 6 months. Households had on average 0.4 LLIN per household member and only 55.4 percent of the births took place in health facilities. Though this was higher than the south-western average of 40.3 percent in 2011 (UBOS & ICF International, 2012), it was lower than the 2016 south-western average of 70.3 percent (UBOS & ICF, 2018). Some 63.4% of the mothers had exclusively breastfed their youngest child for up to 6 months. Only 16.8% of the households had all under-five children sleeping under a long-lasting mosquito net (LLIN). Furthermore, about 55.4% of the children had been born in a health facility. This was significantly higher than the regional average of 40.3% of children born in health facilities as measured in the 2011 DHS (UBOS & ICF International, 2012). In terms of distribution of respondents by village economies, some 18.1% of our respondents were located in forestry villages while 26.1% were in banana cultivation villages. Around 36.6% of the respondents were located in trading villages while the remaining 19.2% were from villages whose pastoralism was the main economic activity. More descriptive statistics are presented in the table below.

#### 4.2 **Empirical results**

#### 4.2.1 Main model results

The emprical results are presented in Table 3 below. In order to make the appropriate comparisons, results for probit regressions. After ascertaining that the models are well

VARIABLES	Mean	Min	Max	SD
CBHI status	0.438	0	1	0.497
Years in CBHI	5.138	1	11	3.015
Child's age (months)	30.202	5.550	60.580	15.152
Mother's age (years)	30.204	14.010	56.540	7.164
Child is male	0.481	0	1	0.500
Birthweight	3.186	2.0	5.6	0.529
LLIN per capita	0.436	0	3	0.281
Exclusive breastfeeding	0.634	0	1	0.482
Health facility delivery	0.554	0	1	0.498
Catholic	0.504	0	1	0.501
Parental (some) secondary education	0.304	0	1	0.460
Wealth index				
Quintile 1	-1.280	-1.754	-0.999	0.193
Quintile 2	-0.784	-0.997	-0.565	0.123
Quintile 3	-0.286	-0.564	-0.052	0.153
Quintile 4	0.315	-0.051	0.766	0.254
Quintile 5	2.212	0.767	8.365	1.435
Food adequacy	0.534	0	1	0.499
Household diet diversity score	4.080	0	8	1.280
Husband's employment - casual labour	0.356	0	1	0.479
Mother's employment -casual labour	0.101	0	1	0.302
Household size <4	0.401	0	1	0.491
Proportion of under five	0.265	0.077	0.667	0.129
Access to information	-0.000	-2.631	4.472	1.289
Neighbour in CBHI	0.692	0	1	0.462
Waiting time	88.621	5	540	108.851
No of groups' membership	1.829	0	5	1.003
Groups membership squared	4.351	0	25	4.198
Interface with TBA	0.528	0	1	0.500
No of burial groups in the village	5.601	1	10	3.349
Distance to hospital	11.239	5.45	17.40	3.349
No of household in the village	120.412	25	250	57.692
Village has a health centre	0.401	0	1	0.491
Village has a school	0.634	0	1	0.482
Village economy-pastoralism	0.192	0	1	0.394
Village economy-banana cultivation	0.261	0	1	0.440
Village CBHI demand rate	0.482	0	0.75	0.183
Social influence	$-6.79  imes 10^{-09}$	-5.006	2.013	1.506
Size of burial group	71.366	18	200	26.054
Stunting	0.425	0	1	0.495
N		464		

#### Table 2: Descriptive results

Mean for years in CBHI corresponds to only the CBHI participating households (n=203). 44% of the variable birthweight (207 obs) are imputed, see imputation note

fit<sup>4</sup>, results indicate that there was no statistically significant association between CBHI participation and stunting. The main focus is therefore drawn on the average partial effects of the 2SRI- IV results presented in Model 3. After implementing the second stage probit regression of the probability of stunting, it is established that after controlling for child-specific, household, and village level covariates, the probability of stunting reduces by 5.7 percentage points for each year a household is insured, significant at 1 percent.

Household socioeconomic welfare assessed by wealth quintiles was also very important in determining child stunting. It was found that compared with the poorest quintile households, children in households within the second poorest quintile are able to reduce stunting by close to 19.4 percentage points, the third quintile by 14 percentage points and 19.1 percentage points for children within the fourth quintile households. While a statistically significant reduction in stunting for children in fifth quintile households was not observed, a large negative coefficient indicates that any move upwards in the welfare of the household was good for stunting reduction. In addition to household socioeconomic status, children in households of small sizes (of four members or less) were likely to reduce stunting by 11.9 by percentage points compared to children in relatively larger households. The probability of stunting for households living further from health facilities was lower by 3 percentage points.

Looking at the variables that increased the probability of stunting, it was found that an increase in the age of the child by one month was associated with increasing the probability of child stunting increased by 3.2 percentage points. However, the negative sign on the quadratic term indicated that probability of stunting increases in a declining trend and is not significant after about 32-33 months. Stunting increased by 13.6 percentage points for children who had been exclusively breastfed for six months, a strange finding. An increase by 1 in the number of burial groups in the village was associated with increasing child stunting by 3 percentage points. The number of burial groups in a

<sup>&</sup>lt;sup>4</sup>Variance Information Factor (VIF) test for multicollinearity shows a mean VIF of 11.5 which is within the threshold of 30 (O'Brien, 2007; StataCorp, 2015) so we are confident that the models do not violate conventional multicollinearity standards. Both the linktest (Pregibon, 1980) and the goodness of fit (Hosmer, Lemeshow, & Sturdivant, 2013) show that the model is correctly specified.

	(1)		(2)		(3)	
	Model 1: Pi	obit	Model 2: Pi	obit	Model 3: IV	/-2SRI
	APE	SE	APE	SE	APE	SE
CBHI participation	0.038	(0.064)				
Years in CBHI		. ,	0.006	(0.009)	-0.057 * * *	(0.022)
Child's age (months)	0.031***	(0.006)	0.031***	(0.006)	0.032***	(0.007)
Child's age square	-0.000 * * *	(0.000)	-0.000 * * *	(0.000)	-0.000 * * *	(0.000)
Mother's age (Base: <24.9 years)						
25 - 34.9 years	-0.036	(0.056)	-0.042	(0.056)	0.027	(0.067)
35 - above	0.023	(0.073)	0.014	(0.073)	0.110	(0.086)
Child is male	0.022	(0.043)	0.023	(0.043)	0.028	(0.048)
Birthweight	-0.058	(0.042)	-0.060	(0.042)	-0.068	(0.046)
Health facility delivery v0.014	(0.048)	0.015	(0.048)	0.014	(0.053)	
LLIN per capita	0.024	(0.082)	0.023	(0.082)	0.031	(0.100)
Exclusive breastfeeding	0.121***	(0.045)	0.121***	(0.044)	0.136***	(0.050)
Catholic	-0.062	(0.048)	-0.062	(0.048)	-0.054	(0.052)
Parental (some) secondary education	-0.071	(0.055)	-0.074	(0.054)	-0.077	(0.061)
Wealth index (Base: quintile 1)						
Quintile 2	-0.190 * * *	(0.069)	-0.187 * * *	(0.069)	-0.193 * *	(0.079)
Quintile 3	-0.177 * *	(0.070)	-0.177 * *	(0.070)	-0.140*	(0.081)
Quintile 4	-0.220 * * *	(0.072)	-0.218 * * *	(0.072)	-0.191 * *	(0.084)
Quintile 5	-0.204 * *	(0.084)	-0.201 **	(0.084)	-0.152	(0.098)
Food adequacy	0.080*	(0.048)	0.080*	(0.048)	0.072	(0.056)
Household diet diversity score	-0.015	(0.019)	-0.014	(0.019)	-0.015	(0.022)
Husband employment - casual labour	0.013	(0.051)	0.014	(0.051)	0.019	(0.059)
Wife employment - casual labour	0.010	(0.073)	0.007	(0.072)	0.009	(0.084)
Household size <4	-0.135 * *	(0.055)	-0.137 * *	(0.055)	-0.119*	(0.062)
Proportion of under five	0.203	(0.218)	0.217	(0.219)	0.093	(0.250)
Access to information	-0.020	(0.021)	-0.021	(0.021)	-0.020	(0.024)
Neighbour in CBHI	0.009	(0.056)	0.009	(0.056)	0.036	(0.066)
Waiting time	-0.000	(0.000)	-0.000	(0.000)	-0.000	(0.000)
No of group memberships	0.000	(0.029)	0.002	(0.028)	0.050	(0.035)
Interface with TBA	-0.037	(0.050)	-0.039	(0.050)	-0.043	(0.057)
No of burial groups in village	0.010	(0.012)	0.010	(0.012)	0.029**	(0.014)
Distance to hospital	-0.014	(0.016)	-0.014	(0.016)	-0.031*	(0.018)
No of household in village	0.000	(0.001)	0.000	(0.001)	0.000	(0.001)
Village has a health centre	-0.091	(0.096)	-0.089	(0.097)	-0.118	(0.110)
Village has a school	0.040	(0.075)	0.040	(0.075)	0.050	(0.087)
Village economy-pastoralism	-0.085	(0.102)	-0.085	(0.101)	0.010	(0.115)
Village economy-banana cultivation	0.006	(0.066)	0.008	(0.065)	0.038	(0.074)
Residuals					0.077***	(0.024)
Link test (hat squared)	-0.111	(0.185)	-0.120	(0.184)		
Mean VIF (uncentered)	11.55	. ,	11.51	. ,		
Goodness of fit chi square	0.1710		0.5278			
Pseudo R-squared	0.1226		0.1227		0.1402	7
First stage Wald test					24.678	3
N	464		464		464	

## Table 3: Impact of CBHI on stunting: Average Partial Effects

Robust standard errors in parenthesis for Models 1 & 2.and Bootstrapped standard errors (1000 replications) for Model 3 \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

village indicates the depth of traditional informal social support strategies which in turn limits households' propensity for formal insurance.

To further elucidate on the effect on an extra year of CBHI participation, we plot predictive marginal probabilities for each year in CBHI. Results plotted in Figure 4 below indicate that the probability of stunting reduced consistently as households spend more years in CBHI. Specifically, it reduced from 51.5 percent for children in households with no CBHI participation to 37.6 percent in households with 5 years of consistent participation and remained statistically significant until seven years in CBHI. However, after five years, body mass index rather than stunting becomes the appropriate nutritional indicator. In addition, other households and individual child determinants influence the overall health status of a child. Nonetheless, the reduction in stunting, causally associated with CBHI participation is very clear.





One unexpected finding in these results is the size and sign of the coefficients of exclusive breastfeeding. As established, exclusive breastfeeding for at least six months of a child's life, is essential for child health through growth and nutrition (Kamudoni,

Maleta, Shi, & Holmboe-Ottesen, 2015; Kuchenbecker et al., 2015) and therefore should reduce stunting too. We therefore expect a negative sign on the coefficient on this variable. However, we get a positive and statistically significant coefficient. We suspect that there might be measurement error through social desirability bias (Martinelli & Parker, 2009; Meshnick, 2015) or recall error, (Aarts et al., 2000; Bland, Rollins, Solarsh, Van den Broeck, & Coovadia, 2003; Gillespie, D'Arcy, Schwartz, Bobo, & Foxman, 2006) especially when a mother has had more children (Cupul-Uicab, Gladen, Hernández-Ávila, & Longnecker, 2009), as is the case in most developing countries like Uganda. The worst case scenario is that it influences the result that we get. We therefore omit the variable from the model and the results remain stable.

#### 4.2.2 Some heterogeneous treatment effects

#### **Boys and girls:**

To understand how the effect of CBHI is distributed across boy and girls, we undertake separate outcome regressions. Whilst both boys and girls have negative coefficients, most of the stunting reduction was in boys. In particular, an extra year of participating in CBHI was associated with a 7.2 percentage point reduction in stunting among boys.

Moreover, the coefficient of stunting reduction among girls was not statistically significant and the coefficient of the residuals is not significant, indicating that endogeneity was controlled. Summary results showed that there was no significant difference between stunting prevalence for boys (43.0 percent) and girls (41.9 percent) therefore it is clear that the causal effect is skewed more to boys than girls. In addition, the effect of household socioeconomic status was strongest in the boys sub-sample only. Compared to boys in quintile 1, the probability of stunting for boys in quintiles 2, 3 and 4 was lower by 36, 28 and 36 percentage points respectively.

	Model 1: Boys Model 2: 0		Model 2: Gi	Girls	
	Coeff	SE	Coeff	SE	
Years in CBHI	-0.0722*	(0.0396)	-0.0462	(0.0417)	
Child's age (months)	0.0280**	(0.0129)	0.0376***	(0.0119)	
Child's age square	-0.0004 * *	(0.0002)	-0.0006 * * *	(0.0002)	
Exclusive breastfeeding	0.1589*	(0.0916)	0.1388*	(0.0794)	
Catholic	-0.0698	(0.0962)	-0.0142	(0.0870)	
Wealth index (quintile 1)					
Quintile 2	-0.3643 * *	(0.1445)	0.0082	(0.1285)	
Quintile 3	-0.2705 **	(0.1350)	-0.0061	(0.1392)	
Quintile 4	-0.3601 * *	(0.1504)	-0.0679	(0.1370)	
Quintile 5	-0.2209	(0.1796)	-0.0844	(0.1556)	
Food adequacy	0.1763*	(0.0966)	-0.0149	(0.0914)	
Household diet diversity score	-0.0655*	(0.0391)	0.0130	(0.0372)	
Household size <=4	0.0139	(0.1221)	-0.1861*	(0.1004)	
No of burial groups in village	0.0499*	(0.0291)	0.0140	(0.0254)	
Residuals	0.0831*	(0.0431)	0.0721	(0.0460)	
All other covariates	(YES)		(YES)		
Ν	(223)		(241)		

Table 4: Effect of CBHI membership on stunting between boys and girls

Bootstrapped standard errors (1000 replications) \*\*\* p<0.01, \*\* p<0.05, \* p<0.1

#### Local Average Treatment Effects

Imbens and Angrist (1994) suggested a precise measure of treatment effects, the local average treatment effect, which under mild assumptions, estimates the treatment effect on only the subjects who receive the treatment. To estimate this, the outcome regression is undertaken on only the subsample of CBHI participant households. For these households, one extra year of CBHI participation was associated with reducing the probability of stunting by 14.8 percentage points, almost three times the effect on the whole population.

	Coeff	SE
Years in CBHI	-0.1488 * *	(0.0694)
Child's age (months)	0.0279*	(0.0155)
Child's age square	-0.0005*	(0.0002)
Mother's age (<24.9 years)		
25 - 34.9 years	0.3202***	(0.1198)
35 - above	0.4490 * * *	(0.1585)
Birthweight	-0.1683*	(0.0945)
Exclusive breastfeeding	0.2390**	(0.1076)
Wealth index (Quintile 1)		
Quintile 2	-0.3432 **	(0.1741)
Quintile 3	-0.0857	(0.1919)
Quintile 4	-0.1519	(0.1845)
Quintile 5	-0.1024	(0.2332)
No of burial groups in village	0.0763**	(0.0378)
Residuals	0.1520**	(0.0688)
All other covariates	(YES)	
N	(203)	

 Table 5: Local average treatment effects

Bootstrapped Standard errors (1000 replications) \*\*\*p<0.01, \*\*p<0.05, \*p<0.1

#### 4.3 Robustness checks

Even when insurance uptake has increased over the years, only 43.8% of our sample were insured. The implication of this is that we have a large number of zeros in our data and hence there is over dispersion in our data. For robustness checks we therefore elicit to use alternative models which are consistent with count data with large number of zeros. Normally, count data is analysed with Poisson models, however due to substantial number of zeros in the data, zero inflated models such as Zero-inflated Poisson and Zero-inflated negative binomial models have been developed to deal with this kind of distribution, zero or non-zero and over dispersion of the count variables (Cameron & Trivedi, 2009, Chap 17). To compare with our main model, (which following Terza (2017) also uses Generalised Linear Model (GLM) probit specifications) we estimate a zero-inflated Poisson and zero-inflated negative binomial model shows being the specifications.

with highly dispersed data (Hu, Pavlicova, & Nunes, 2011). Comparing the results, we find that both alternative models produce similar results and account for endogeneity with similar level of efficiency. Our main coefficient is however 14 percent larger than the comparative models. Nonetheless, we are confident that even using different but suitable specifications, the similar results can be arrived at.

(1)(2)(3)Main First stage First stage Model ZIP model ZINB model -0.057 \* \* \*Years in CBHI -0.049 \* \* \*(0.022)-0.049 \* \*(0.021)(0.019)Residuals 0.069\*\*\* 0.069\*\*\* 0.077\*\*\* (0.024)(0.023)(0.020)All other covariates YES YES YES Ν 464 464 464

**Table 6:** Robustness checks with alternative specifications

Bootstrapped standard errors (1000 replications) in parentheses. \*\*\*p<0.01, \*\*p<0.05, \*p<0.1

# 5 Discussion and conclusion

The main results indicate that each year a household participated in insurance was associated wit reducing the probability of child stunting by 5.7 percentage points. Our data does not give us an opportunity to conclusively elucidate how this effect happens. However, we postulate three main pathways through which this effect might happen. The first is through social learning and social network effects associated with funeral groups through which households access insurance. Broadly, traditional funeral groups in south-western Uganda are meant for the primary purpose of funeral insurance but also perform other functions such as providing lending and borrowing services to their members at different extents. In extension to these services, funeral groups also provide a platform for infusing health information and facilitate behaviour change. Previous research on this front has primarily been focused on controlling of river blindness in the region and the authors find that not only was it cheaper to deliver health information to communities (Katabarwa, Mutabazi, & Richards, 1999) but also communities in which these platforms were used showed high rate of information take-up and more prevention than those in which such groups were used (Katabarwa, Habomugisha, Agunyo, et al., 2010; Katabarwa, Habomugisha, Eyamba, Agunyo, & Mentou, 2010; Katabarwa et al., 2015). However, at this stage we do not differentiate the funeral groups. The only difference is that some groups choose to participate in insurance and others do not. Indeed there could be some self-selection issues which out data cannot disentangle. However, what we observe is that funeral groups that participated in insurance had greater cohesion, evidenced by the number of times the met in month (2.4 times) compared to those that were not in insurance (2.1 times), a difference of 14 percent. This can also be appraised in the same way as intensive exposure to health information. Such has been found to have a significant effect on health behaviours such as adoption of long-lasting mosquito nets (Kilian et al., 2016; MacIntyre et al., 2012). In a related dimension, there might be social learning happening within the groups. These funeral groups are built on a strong cohesion of kin associations where behaviours and norms are established and in some instances, diversion from norms attracts reprimand (Katabarwa, 1999). We believe that in these highly socially controlled groups, positive learning happens when healthier households teach unhealthy households for instance on nutrition and hygiene. Our postulation here is supported by research in India that shows that households learn from each other on the adoption of improved sanitation and hygiene strategies (Shakya, Christakis, & Fowler, 2014, 2015).

Another channels through which these effects might come about is through service utilisation. Many studies have found that health insurance coverage greatly facilitates utilisation of health services for mothers (?) and for children (Gajate-Garrido & Ahiadeke, 2015; Singh et al., 2015). Findings from this study are in line with this literature. For instance, CBHI participating mothers were more likely to have four or more antenatal care (ANC) visits, receive the essential ANC services and have a postnatal care visit compared to non-participating mothers. Moreover, when asked if a child had experienced sickness conditions manifesting in a cough, or fever or diarrhoea in fourteen days prior to the survey, only 3.5 percent of children in insured households reported the affirmative compared to 6.1 percent of the children in uninsured households. These differences give a compelling correlation through which health services utilisation can translate into health improvements.

	(1) Overall Mean	(2) Mean CBHI	(3) Mean Non CBHI	(4) Mean difference	(5) t- statistic
Attended at least 4 ANC visits	71.55	89.16	57.85	-0.313 * * *	-7.88
Attended at least 1 postnatal visit	82.76	84.73	81.24	-0.0350	-0.99
Received essential ANC services	40.73	50.25	33.33	-0.169 * * *	-3.73
Reported the 3 child illnesses	04.96	03.45	06.13	0.0268	1.32
N	464	203	261	464	

 Table 7: Differences in services utilisation as pathways of impact

Significance of the mean difference reported for p < 0.1, p < 0.05, p < 0.01

Finally, out data is strongly suggestive of financial protection, which would facilitate savings and consumption. It has been found elsewhere that health insurance is associated with an increase in households' consumption of both food and non-food goods (Bai & Wu, 2014; Wagstaff & Pradhan, 2005). These findings would be in line with a large body of literature that shows that health insurance provided considerable financial protection. To test this in our data, we ran OLS regressions on the association between CBHI participation and household income and also between CBHI participation and reported health expenditure. The results indicated a significant negative association between CBHI participation and health expenditure, implying a possibility of financial protection. In this way, households would be able to release extra resources for food and non-food consumption that could have a positive effect on child health.

VARIABLES	(1) CBHI Participation	(2) Cost of Participation (ugx)			
Household income (log)	-0.0145 (0.0697)				
CBHI participation	<b>、</b>	-25,273**			
		(10,977)			
Constant	-0.0963	55,759***			
	(0.749)	(9,018)			
Observations	464	304			
R-squared		0.018			
Robust standard errors in parentheses.					

**Table 8:** Association of household income and CBHI participation and CBHI participation and cost of care

Robust standard errors in parentheses. \*\*\* p < 0.01, \*\* p < 0.05, \* p < 0.1

# 6 Conclusion

In conclusion, after applying a novel IV approach that follows Terza (2017) to account for the endogeneity of selection in CBHI and child stunting, we found CBHI reduced the probability of stunting by up to 5.7 percentage points for each year a household participated in CBHI. This implies that children who are born in households participating in CBHI and remain so until the children are five have a 28.5 percentage point lower probability of stunting at five years. In a country where about one-third of all underfive children were stunted, such reductions were profound. Current and future policy discussions should consider promoting this kind of insurance in rural areas not only to mobilise more resources for the health systems but also for its contributions to improving child health. However, it was found that a large part of this effect was dominated by boys over girls. This raises the concern and the requirement to focus more on girls and women who do not often have equitable access to health services and yet need them often the most. The pathways of impact discussed here at at best, speculative, nonetheless, plausible. There are a couple of outstanding questions which future research might tackle. For instance, how groups function and how social control might enhance positive behaviour change. Moreover, it will be important to gather more evidence on the health

outcome impacts of health insurance so that these can be put into consideration when CBHI and other kinds of health insurance are promoted to target populations.

# References

- Aarts, C., Kylberg, E., Hornell, A., Hofvander, Y., Gebre-Medhin, M., & Greiner, T. (2000). How exclusive is exclusive breastfeeding? A comparison of data since birth with current status data. *International Journal of Epidemiology*, 29(6), 1041–1046.
- Acharya, A., Vellakkal, S., Taylor, F., Masset, E., Satija, A., Burke, M., & Ebrahim, S. (2013). The Impact of Health Insurance Schemes for the Informal Sector in Low-and Middle-Income Countries: A Systematic Review. *The World Bank Research Observer*, 28(2), 236–266.
- Adebayo, E. F., Uthman, O. A., Wiysonge, C. S., Stern, E. A., Lamont, K. T., & Ataguba, J. E. (2015). A systematic review of factors that affect uptake of community-based health insurance in low-income and middle-income countries. *BMC Health Services Research*, 15, 543.
- Angrist, J. D., & Imbens, G. W. (1995). Two-Stage Least Squares Estimation of Average Causal Effects in Models With Variable Treatment Intensity. *Journal of the American Statistical Association*, 90(430), 431–442.
- Angrist, J. D., Imbens, G. W., & Rubin, D. B. (1996). Identification of Causal Effects Using Instrumental Variables. *Journal of the American Statistical Association*, 91(434), 444–455.
- Asfaw, A., & von Braun, J. (2004a). Can community health insurance schemes shield the poor against the downside health effects of economic reforms? The case of rural Ethiopia. *Health Policy*, 70(1), 97–108.
- Asfaw, A., & von Braun, J. (2004b). Is Consumption Insured against Illness? Evidence on Vulnerability of Households to Health Shocks in Rural Ethiopia. *Economic Development and Cultural Change*, 53(1), 115–129.
- Atinga, R. A., Abiiro, G. A., & Kuganab-Lem, R. B. (2015). Factors influencing the decision to drop out of health insurance enrolment among urban slum dwellers in Ghana. *Tropical Medicine and International Health*, 20(3), 312–321.
- Bai, C. E., & Wu, B. (2014). Health insurance and consumption: Evidence from China's New Cooperative Medical Scheme. *Journal of Comparative Economics*, 42(2), 450–469.
- Bennett, S., Creese, A., & Monasch, R. (1998). *Health Insurance Schemes for People Outside Formal Sector Employment* (Tech. Rep.). Geneva: World Health Organisation.
- Bland, R. M., Rollins, N. C., Solarsh, G., Van den Broeck, J., & Coovadia, H. M. (2003). Maternal recall of exclusive breast feeding duration. *Archives of Disease in Childhood*, 88, 778–783.
- Bonfrer, I., Van De Poel, E., Gustafsson-Wright, E., & van Doorslaer, E. (2015). *Effects of a subsidized voluntary health insurance on insured and uninsured in Nigeria* (No. April).
- Cai, B., Small, D. S., & Have, T. R. T. (2011). Two-stage instrumental variable methods for estimating the causal odds ratio: Analysis of bias. *Statistics in Medicine*, *30*(15), 1809–1824.
- Cameron, A. C., & Trivedi, P. K. (2009). *Microeconometrics using Stata*. Stata Press.
- Carpenter, J. P. (2007). Punishing free-riders: How group size affects mutual monitoring and the provision of public goods. *Games and Economic Behavior*, 60(1), 31–51.

- Carrin, G., Waelkens, M.-P., & Criel, B. (2005). Community-based health insurance in developing countries: a study of its contribution to the performance of health financing systems. *Tropical Medicine and International Health*, 10(8), 799–811.
- Case, A., & Paxson, C. (2010). Causes and Consequences of Early-Life Health. *Demography*, 47(0), S65–S85.
- Chao, J. C., & Swanson, N. R. (2005). Consistent Estimation with a Large Number of Weak Instruments. *Econometrica*, 73(5), 1673–1692.
- Chatterji, P. (2006). Illicit drug use and educational attainment. *Health Economics*, 15, 489–511.
- Criel, B., Atim, C., Basaza, R., Blaise, P., & Waelkens, M. P. (2004). Editorial: Community health insurance (CHI) in sub-Saharan Africa: researching the context. *Tropical Medicine and International Health*, 9(10), 1041–1043.
- Cupul-Uicab, L. A., Gladen, B. C., Hernández-Ávila, M., & Longnecker, M. P. (2009). Reliability of reported breastfeeding duration among reproductive-aged women from Mexico. *Maternal and Child Nutrition*, 5(2), 125–137.
- De Allegri, M., Kouyaté, B., Becher, H., Gbangou, A., Pokhrel, S., & Sanon, M. (2006). Understanding enrolment in community health insurance in sub-Saharan Africa : a population-based case - control study in rural Burkina Faso. *Bulletin of the World Health Organization*, 84(11), 852–858.
- De Weerdt, J., & Dercon, S. (2006). Risk-sharing networks and insurance against illness. *Journal of Development Economics*, 81(2), 337–356.
- De Weerdt, J., Dercon, S., Bold, T., & Pankhurst, A. (2007). Membership Based Indigenous Insurance Associations in Ethiopia and Tanzania. In M. Chen, R. Jhabvala, R. Kanbur, & C. Richards (Eds.), *Membership based organizations of the poor* (pp. 157–176). Routledge.
- Dekker, M., & Wilms, A. (2010, mar). Health Insurance and Other Risk-Coping Strategies in Uganda: The Case of Microcare Insurance Ltd. *World Development*, *38*(3), 369– 378.
- Dercon, S., De Weerdt, J., Bold, T., & Pankhurst, A. (2006). Group-based funeral insurance in Ethiopia and Tanzania. *World Development*, *34*(4), 685–703.
- Dercon, S., Vargas, R., Clarke, D., Outes-Leon, I., & Taffesse, S. A. (2014). Offering rainfall insurance to informal insurance groups : Evidence from a field experiment in Ethiopia. *Journal of Development Economics*, *106*, 132–143.
- Dewey, K. G., & Begum, K. (2011). Long-term consequences of stunting in early life. *Maternal and Child Nutrition*, 7(SUPPL. 3), 5–18.
- Dow, W. H., & Schmeer, K. K. (2003). Health insurance and child mortality in Costa Rica. *Social Science & Medicine*, 57, 975–986.
- Dror, D. M., Shahed Hossain, S. A., Majumdar, A., Koehlmoos, T. L. P., John, D., & Panda, P. K. (2016). What factors affect voluntary uptake of community-based health insurance schemes in low- and middle-income countries? A systematic review and meta-analysis. *PLoS ONE*, *11*(8), e0160479.
- Duggan, M. B. (2010). Anthropometry as a tool for measuring malnutrition: impact of the new WHO growth standards and reference. *Annals of Tropical Peadiatrics*, *30*(1), 1–17.

- Ekman, B. (2004). Community-based health insurance in low-income countries: A systematic review of the evidence. *Health Policy and Planning*, 19(5), 249–270.
- Fafchamps, M., & La Ferrara, E. (2012). Self-Help Groups and Mutual Assistance: Evidence from Urban Kenya. *Economic Development and Cultural Change*, 60(4), 707– 733.
- Fink, G., Robyn, P. J., Sié, A., & Sauerborn, R. (2013). Does health insurance improve health? Evidence from a randomized community-based insurance rollout in rural Burkina Faso. *Journal of Health Economics*, 32(6), 1043–1056.
- Gajate-Garrido, G., & Ahiadeke, C. (2015). *The effect of insurance enrollment on maternal and child health care utilization: The case of Ghana.* Washington DC.
- Genicot, G., & Ray, D. (2003). Group Formation in Risk-Sharing Arrangement. *The Review of Economic Studies*, 70(1), 87–113.
- Gibson, T. B., Song, X., Alemayehu, B., Wang, S. S., Waddell, J. L., Bouchard, J. J. R., & Forma, F. (2010). Cost sharing, adherence, and health outcomes in patients with diabetes. *The American Journal of Managed Care*, 16(8), 589–600.
- Gillespie, B., D'Arcy, H., Schwartz, K., Bobo, J., & Foxman, B. (2006). Recall of age of weaning and other breastfeeding variables. *International Breastfeeding Journal*, 1(1), 4.
- Glewwe, P., Jacoby, H. G., & King, E. M. (2001). Early childhood nutrition and academic achievement: A longitudinal analysis. *Journal of Public Economics*, *81*(3), 345–368.
- Government of Rwanda. (2016). GOVERNING THE ORGANISATION, FUNC-TIONING AND MANAGEMENT OF HEALTH INSURANCE SCHEMES IN RWANDA. Retrieved from http://www.moh.gov.rw/fileadmin/templates/HLaws/ Health{\_}Insurance{\_}law{\_}2016.pdf
- Hansen, C., Hausman, J., & Newey, W. (2008). Estimation With Many Instrumental Variables. *Journal of Business & Economic Statistics*, 26(4), 398–422.
- Hendriks, M. E., Rosendaal, N. T. A., Wit, F. W. N. M., Bolarinwa, O. A., Kramer, B., Brals, D., ... Schultsz, C. (2016, jan). Sustained effect of health insurance and facility quality improvement on blood pressure in adults with hypertension in Nigeria: A population-based study. *International Journal of Cardiology*, 202, 477–484.
- Hendriks, M. E., Wit, F. W. N. M., Akande, T. M., Kramer, B., Osagbemi, G. K., Tanovic, Z., ... Schultsz, C. (2014, apr). Effect of Health Insurance and Facility Quality Improvement on Blood Pressure in Adults With Hypertension in Nigeria: A Population-Based Study. *JAMA Internal Medicine*, 24(4), 555–563.
- Hosmer, D. W., Lemeshow, S., & Sturdivant, R. X. (2013). *Applied Logistic Regression* (3rd ed.). John Wiley & Sons.
- Hu, M.-C., Pavlicova, M., & Nunes, E. V. (2011). Zero-Inflated and Hurdle Models of Count Data with Extra Zeros: Examples from an HIV-Risk Reduction Intervention Trial. *The American Journal of Drug and Alcohol Abuse*, 37(5), 367–375.
- Imbens, G. W., & Angrist, J. D. (1994). Identification and Estimation of Local Average Treatment Effects. *Econometrica*, 62(2), 467–475.
- Jehu-Appiah, C., Aryeetey, G., Agyepong, I., Spaan, E., & Baltussen, R. (2012). Household perceptions and their implications for enrolment in the National Health Insurance Scheme in Ghana. *Health Policy and Planning*, 27(3), 222–233.

- Jung, J., & Streeter, L. J. (2015). Does health insurance decrease health expenditure risk in developing countries? The case of China. *Southern Economic Journal*, *82*(2), 361–384.
- Jutting, J. P. (2004). Do community-based health insurance schemes improve poor people's access to health care? Evidence from rural senegal. *World Development*, 32(2), 273–288.
- Kamudoni, P., Maleta, K., Shi, Z., & Holmboe-Ottesen, G. (2015). Exclusive breastfeeding duration during the first 6 months of life is positively associated with length-forage among infants 6-12 months old, in Mangochi district, Malawi. *European Journal of cCinical Nutrition*, *69*(1), 96–101.
- Katabarwa, M. (1999). Modern health services versus traditional engozi system in Uganda. *The Lancet*, 354, 343.
- Katabarwa, M., Habomugisha, P., Agunyo, S., McKelvey, A. C., Ogweng, N., Kwebiiha, S., ... McFarland, D. (2010). Traditional kinship system enhanced classic community-directed treatment with ivermectin (CDTI) for onchocerciasis control in Uganda. *Transactions of the Royal Society of Tropical Medicine and Hygiene*, 104(4), 265–272.
- Katabarwa, M., Habomugisha, P., Eyamba, A., Agunyo, S., & Mentou, C. (2010). Monitoring ivermectin distributors involved in integrated health care services through community-directed interventions - A comparison of Cameroon and Uganda experiences over a period of three years (2004-2006). *Tropical Medicine and International Health*, 15(2), 216–223.
- Katabarwa, M., Habomugisha, P., Eyamba, A., Byamukama, E., Nwane, P., Arinaitwe, A., ... Khainza, A. (2015). Community-directed interventions are practical and effective in low-resource communities: Experience of ivermectin treatment for on-chocerciasis control in Cameroon and Uganda, 2004-2010. *International Health*, *8*(2), 116–123.
- Katabarwa, M., Mutabazi, D., & Richards, F. (1999). The community-directed, ivermectin-treatment programme for onchocerciasis control in Uganda An evaluative study (1993-1997). *Annals of Tropical Medicine and Parasitology*, 93(7), 727–735.
- Kilian, A., Lawford, H., Ujuju, C. N., Abeku, T. A., Nwokolo, E., Okoh, F., & Baba, E. (2016). The impact of behaviour change communication on the use of insecticide treated nets: a secondary analysis of ten post-campaign surveys from Nigeria. *Malaria Journal*, 15(1), 422.
- Kuchenbecker, J., Jordan, I., Reinbott, A., Herrmann, J., Jeremias, T., Kennedy, G., ... Krawinkel, M. B. (2015). Exclusive breastfeeding and its effect on growth of Malawian infants: results from a cross-sectional study. *Paediatrics and International Child Health*, 35(1), 14–23.
- Landmann, A., & Frölich, M. (2015). Can health-insurance help prevent child labor? An impact evaluation from Pakistan. *Journal of Health Economics*, *39*, 51–59.
- Lei, X., & Lin, W. (2009). The New Cooperative Medical Scheme in Rural China: Does More Coverage Mean More Service and Better Health? *Health Economics*, 46, 25– 46.
- Levy, H., & Meltzer, D. (2004). What do we really know about whether health insurance affects health? In C. G. McLaughlin (Ed.), *Health policy and the uninsured* (pp. 179–204). Washington DC: The Urban Institute Press.

- Levy, H., & Meltzer, D. (2008). The impact of health insurance on health. *Annual Review* of *Public Health*, 29, 399–409.
- Lu, C., Mejia-Guevara, I., Hill, K., Farmer, P., Subramanian, S. V., & Binagwaho, A. (2016). Community-based health financing and child stunting in rural Rwanda. *American Journal of Public Health*, 106(1), 49–55.
- MacIntyre, K., Littrell, M., Keating, J., Hamainza, B., Miller, J., & Eisele, T. P. (2012). Determinants of hanging and use of ITNs in the context of near universal coverage in Zambia. *Health Policy and Planning*, 27(4), 316–325.
- Martinelli, C., & Parker, S. (2009). Deception and misreporting in a social program. *Journal of the European Economic Association*, 7(4), 886–908.
- Mebratie, A., Sparrow, R., Alemu, G., & Bedi, A. S. (2013). Community-Based Health Insurance Schemes: A Systematic Review.
- Meshnick, S. R. (2015). Editorial: Malaria control: Tortoises and hares. *American Journal* of Tropical Medicine and Hygiene, 92(5), 885–886.
- Mullahy, J. (1997). Instrumental-Variable Estimation of Count Data Models: Applications to Models of Cigarette Smoking Behavior. *The Review of Economics and Statistics*, 79(4), 586–593.
- Mullahy, J. (1998). Much ado about two: Reconsidering retransformation and the twopart model in health econometrics. *Journal of Health Economics*, 17(3), 247–281.
- Musau, S. N. (1999). *Community Based Health Insurance: Experiences and Lessons learned from East Africa* (Vol. 34; Tech. Rep. No. 34). Bethseda MD: ABT Associates.
- Namugumya, B., Sethuraman, K., Sommerfelt, E. A., Oot, L., Kovach, T., & Musimenta, B. (2014). *Reducing Malnutrition in Uganda : Estimates to Support Nutrition Advocacy Uganda PROFILES 2013* (Tech. Rep.). Washington DC; Kampala: FHI 360/ FANTA and Office of the Prime Minister, Uganda.
- Nguyen, C. V. (2012). The Impact of Voluntary Health Insurance on Health Care Utilization and Out-Of-Pocket Payments: New Evidence from Vietnam. *Health economics*, 21, 946–966.
- Nguyen, H. T. H., Rajkotia, Y., & Wang, H. (2011). The financial protection effect of Ghana National Health Insurance Scheme: evidence from a study in two rural districts. *International Journal For Equity in Health*, 10(1), 4.
- Nyandekwe, M., Nzayirambaho, M., & Kakoma, J. B. (2014). Universal health coverage in Rwanda: Dream or reality. *Pan African Medical Journal*, 17.
- O'Brien, R. M. (2007). A caution regarding rules of thumb for variance inflation factors. *Quality and Quantity*, 41(5), 673–690.
- Pan, J., Lei, X., & Liu, G. G. (2016). Health Insurance and Health Status: Exploring a Causal Effect from A Policy Evaluation. *Health Economics*, 25(11), 1389–1402.
- Panda, P., Chakraborty, A., Dror, D. M., & Bedi, A. S. (2014). Enrolment in communitybased health insurance schemes in rural Bihar and Uttar Pradesh, India. *Health Policy and Planning*, 29(8), 960–974.
- Pizer, S. D. (2009). An intuitive review of methods for observational studies of comparative effectiveness. *Health Services and Outcomes Research Methodology*, *9*(1), 54–68.
- Pregibon, D. (1980). Goodness of Link Tests for Generalized Linear Models. *Journal of the Royal Statistical Society. Series C (Applied Statistics)*, 29(1), 15–23.

- Prendergast, A. J., & Humphrey, J. H. (2014). The stunting syndrome in developing countries. *Paediatrics and International Child Health*, 34(4), 250–65.
- Quimbo, S. A., Peabody, J. W., Shimkhada, R., Florentino, J., & Solon, O. (2011, may). Evidence of a causal link between health outcomes, insurance coverage, and a policy to expand access: Experimental data from children in the Philippines. *Health Economics*, 20(5), 620–630.
- Reshmi, B., Sreekumaran, N., & Unnikrishnan, B. (2016). A systematic review of community-based health insurance programs in South Asia. *The International Journal of Health Planning and Management*.
- Roodman, D. (2009). Practitioners' corner: A note on the theme of too many instruments. *Oxford Bulletin of Economics and Statistics*, 71(1), 135–158.
- Schoeps, A., Lietz, H., Sié, A., Savadogo, G., Allegri, M. D., Müller, O., ... Souares, A. (2015). Health insurance and child mortality in rural Burkina Faso. *Global Health Action*, *8*, 1–8.
- Sepehri, A. (2014). How much do I save if I use my health insurance card when seeking outpatient care? Evidence from a low-income country. *Health Policy and Planning*, 29(2), 246–256.
- Sepehri, A., Sarma, S., & Oguzoglu, U. (2011). Does the financial protection of health insurance vary across providers? Vietnam's experience. *Social Science and Medicine*, 73(4), 559–567.
- Shakya, H. B., Christakis, N. A., & Fowler, J. H. (2014). Association between social network communities and health behavior: An observational sociocentric network study of latrine ownership in rural india. *American Journal of Public Health*, 104(5), 930–937.
- Shakya, H. B., Christakis, N. A., & Fowler, J. H. (2015). Social network predictors of latrine ownership. *Social Science and Medicine*, 125, 129–138.
- Singh, K., Osei-Akoto, I., Otchere, F., Sodzi-Tettey, S., Barrington, C., Huang, C., ... Speizer, I. (2015). Ghana's National Health insurance scheme and maternal and child health: a mixed methods study. *BMC Health Services Research*, 15(1), 108.
- Smith, K. V., & Sulzbach, S. (2008). Community-based health insurance and access to maternal health services: Evidence from three West African countries. *Social Science and Medicine*, 66(12), 2460–2473.
- Sood, N., Bendavid, E., Mukherji, A., Wagner, Z., Nagpal, S., & Mullen, P. (2014). Government health insurance for people below poverty line in India: quasi-experimental evaluation of insurance and health outcomes. *BMJ*, *349*, g5114.
- Sood, N., & Wagner, Z. (2016). Impact of health insurance for tertiary care on postoperative outcomes and seeking care for symptoms: quasi-experimental evidence from Karnataka, India. *BMJ open*, *6*(1), e010512.
- Spaan, E., Mathijssen, J., Tromp, N., McBain, F., ten Have, A., & Baltussen, R. (2012). The impact of health insurance in Africa and Asia: a systematic review. *Bulletin of the World Health Organisation*, 90, 685–692.
- Stahl, D. O., & Haruvy, E. (2006). Other-regarding preferences: Egalitarian warm glow, empathy, and group size. *Journal of Economic Behavior and Organization*, *61*(1), 20–41.
- Staiger, D., & Stock, J. (1997). Instrumental Variables Regression with Weak Instruments. *Econometrica*, 65(3), 557–586.

- Stanghellini, E. (2004). Instrumental variables in Gaussian directed acyclic graph models with an unobserved confounder. *Environmetrics*, 15(5), 463–469.
- StataCorp. (2015). TREATMENT-EFFECTS REFERENCE MANUAL: POTENTIAL OUT-COMES / COUNTERFACTUAL OUTCOMES (Stata Statistical Software). College Station, Texas: StataCorp LP.
- Staub, K. E. (2009). Simple Tests for Exogeneity of a Binary Explanatory Variable in Count Data Regression Models. *Communications in Statistics - Simulation and Computation*, 38(9), 1834–1855.
- Stevens, G. A., Finucane, M. M., Paciorek, C. J., Flaxman, S. R., White, R. A., Donner, A. J., & Ezzati, M. (2012). Trends in mild, moderate, and severe stunting and underweight, and progress towards MDG 1 in 141 developing countries: A systematic analysis of population representative data. *The Lancet*, 380(9844), 824–834.
- Stock, J. H., Wright, J. H., & Yogo, M. (2002). A Survey of Weak Instruments and Weak Identification in Generalized Method of Moments. *Journal of Business & Economic Statistics*, 20(4), 518–529.
- Stock, J. H., & Yogo, M. (2005). Testing for Weak Instruments in Linear IV Regression. In D. W. K. Andrews & J. H. Stock (Eds.), *Identification and inference for econometric models: Essays in honor of thomas rothenberg* (pp. 80–108). New York: Cambridge University Press.
- Stukel, T. A., Fisher, E. S., Wennberg, D. E., Alter, D. A., Gottlieb, D. J., & Vermeulen, M. J. (2007). Analysis of Observational Studies in the Presence of Treatment Selection Bias: Effects of Invasive Cardiac Management on AMI Survival Using Propensity Score and Instrumental Variable Methods. *Journal of the American Medical Association*, 297(3), 278 – 285.
- Terza, J. V. (2017). Two-stage residual inclusion estimation: A practitioners guide to Stata implementation. *The Stata Journal*(4), 916–938.
- Terza, J. V., Basu, A., & Rathouz, P. J. (2008). Two-stage residual inclusion estimation: Addressing endogeneity in health econometric modeling. *Journal of Health Economics*, 27(3), 531–543.
- Textor, J., van der Zander, B., Gilthorpe, M. S., Liśkiewicz, M., & Ellison, G. T. (2017). Robust causal inference using directed acyclic graphs: the R package 'dagitty'. *International Journal of Epidemiology*(January), dyw341.
- Thornton, R. L., Hatt, L. E., Field, E. M., Islam, M., Solis-Diaz, F., & Azucena-Gonzalez, M. (2010). Social Security Health Insurance for the Informal Sector in Nicaragua: A Randomized Experiment. *Health economics*, 19, 181–206.
- Twikirize, J. M., & O'Brien, C. (2012). Why Ugandan rural households are opting to pay community health insurance rather than use the free healthcare services. *International Journal of Social Welfare*, 21(1), 66–78.
- UBOS. (2014). *Uganda National Household Survey* 2012/2013 (Tech. Rep.). Kampala: Uganda Bureau of Statistics.
- UBOS. (2017). *Uganda National Household Survey 2016/2017* (Tech. Rep.). Kampala: Uganda Bureau of Statistics.
- UBOS, & ICF. (2018). *Uganda Demographic and Health Survey 2016* (Tech. Rep.). Kampala, Uganda and Rockville Maryland: Uganda Bureau of Statistics and ICF.

- UBOS, & ICF International. (2012). *Uganda Demographic and Health Survey* (Tech. Rep.). Kampala, Uganda; Calverton, Maryland.
- Vogl, T. S. (2014). Height, skills, and labor market outcomes in Mexico. *Journal of Development Economics*, 107, 84–96.
- Wagstaff, A., & Lindelow, M. (2008). Can insurance increase financial risk?. The curious case of health insurance in China. *Journal of Health Economics*, 27(4), 990–1005.
- Wagstaff, A., & Pradhan, M. (2005). *Health Insurance Impacts on Health and Nonmedical Consumption in a Developing Country*. Washington DC.
- Wang, H., & Pielemeier, N. (2012). Community-Based Health Insurance : An Evolutionary Approach to Achieving Universal Coverage in Low-Income Countries. *Journal* of Life Sciences, 6, 320–329.
- Wang, H., Yip, W., Zhang, L., & Hsiao, W. C. (2009). The Impact of Rural Mutual Health Care on Health Status: Evaluation of a Social Experiment in Rural China. *Health Economics*, 82(March), S65–S82.
- Wooldridge, J. M. (2010). *Econometric analysis of cross section and panel data* (Second ed.). Cambridge MA: MIT Press.
- Wooldridge, J. M. (2015). Control Function Methods in Applied Econometrics. *Journal of Human Resources*, 20(2), 420–445.
- Yilma, Z., Mebratie, A., Sparrow, R., Abebaw, D., Dekker, M., Alemu, G., & Bedi, A. S. (2014). Coping with shocks in rural Ethiopia. *The Journal of Development Studies*, 50(7), 1009–1024.
- Yilma, Z., Mebratie, A., Sparrow, R., Dekker, M., Alemu, G., & Bedi, A. S. (2015). Impact of Ethiopia's Community Based Health Insurance on household economic welfare. *World Bank Economic Review*, 29(Supplement), S164–S173.

# Appendix: Supplementary Tables

Years in CBHI	Prob Stunting	Delta-Method Std.Error	Z	P>z	95% Conf. Interval
0	0.531	0.044	11.980	0.000	0.444 - 0.617
1	0.488	0.110	4.440	0.000	0.273 - 0.704
2	0.484	0.121	4.010	0.000	0.247 - 0.720
3	0.344	0.082	4.200	0.000	0.184 - 0.505
4	0.416	0.120	3.460	0.001	0.181 - 0.652
5	0.353	0.091	3.890	0.000	0.175 - 0.531
6	0.147	0.060	2.470	0.014	0.030 - 0.264
7	0.319	0.125	2.540	0.011	0.073 - 0.565
8	0.164	0.115	1.430	0.154	-0.061 - 0.389
9	0.123	0.089	1.380	0.167	-0.051 - 0.296
10	0.133	0.082	1.610	0.108	-0.029 - 0.294
11	0.083	0.059	1.420	0.157	-0.032 - 0.199

 Table 9: Predictive mean probability of stunting at different years of insurance

	(1)		(2)		
	(CBHI Stat	(CBHI Status)		alence)	
	Coeff	se	Coeff	se	
Cluster CBHI demand	0.801	(0.620)	0.628	(0.459)	
Leader & social influence	0.159**	(0.068)	-0.009	(0.043)	
Burial group size	-0.018***	(0.004)	0.001	(0.003)	
Child's age (months)	-0.015	(0.027)	0.090***	(0.020)	
Child's age square	0.000	(0.000)	-0.001 * * *	(0.000)	
Mother's age (<24.9 years)	0.000	(.)	0.000	(.)	
25 - 34.9 years	0.204	(0.211)	-0.121	(0.165)	
35 - above	0.160	(0.275)	0.057	(0.212)	
Child is male	0.147	(0.168)	0.075	(0.127)	
Birthweight	-0.175	(0.170)	-0.171	(0.124)	
Health facility delivery	0.213	(0.186)	0.048	(0.140)	
LLIN per capita	0.156	(0.301)	0.085	(0.239)	
Exclusive breastfeeding	0.207	(0.186)	0.346**	(0.134)	
Catholic	0.249	(0.191)	-0.224	(0.150)	
Parental (some) secondary education	-0.365*	(0.211)	-0.226	(0.161)	
Wealth index (quintile 1)	0.000	(.)	0.000	(.)	
Quintile 2	0.032	(0.252)	-0.520 * *	(0.203)	
Quintile 3	0.374	(0.303)	-0.460 * *	(0.205)	
Quintile 4	0.340	(0.271)	-0.575 * * *	(0.214)	
Quintile 5	0.805**	(0.340)	-0.518 * *	(0.247)	
Food adequacy	0.050	(0.193)	0.209	(0.143)	
Household diet diversity score	0.038	(0.079)	-0.038	(0.056)	
Husband employment -casual labour	0.296	(0.197)	0.041	(0.149)	
Wife employment -casual labour	-0.556*	(0.323)	-0.002	(0.215)	
Household size <4	-0.062	(0.204)	-0.419 * *	(0.165)	
Proportion of under five	0.781	(0.861)	0.631	(0.647)	
Access to information	-0.072	(0.082)	-0.057	(0.063)	
Neighbour in CBHI	0.135	(0.241)	0.009	(0.168)	
Waiting time	0.000	(0.001)	0.001	(0.001)	
No of group memberships	0.825***	(0.132)	0.021	(0.079)	
Interface with TBA	-0.163	(0.190)	-0.185	(0.157)	
No of burial groups in village	0.157***	(0.048)	0.051	(0.036)	
Distance to hospital	-0.210 * * *	(0.061)	-0.029	(0.047)	
No of household in village	-0.001	(0.004)	-0.000	(0.003)	
Village has a health centre	-0.302	(0.365)	-0.245	(0.284)	
Village has a school	0.187	(0.267)	0.100	(0.220)	
Village economy-pastoralism	1.088 * *	(0.503)	-0.149	(0.301)	
Village economy-banana cultivation	0.304	(0.337)	-0.078	(0.233)	
Constant	-0.046	(1.319)	-0.575	(0.886)	
Pseudo R-squared	0.5532		0.1251		
N	464		464		

# Table 10: First Stage and Reduced form Regressions

Bootstrapped standard errors. \* p< 0.1, \*\*  $p_{5}$  0.05, \*\*\* p < 0.01